Author's response to reviews

Title: Respective effects of blood spoliation and erythropoietin treatment on the need for blood transfusion in very premature infants

Authors:

Odile Becquet (odile.becquet@gmail.com)
Delphine Guyot (odile.becquet@gmail.com)
Philippe Kuo (odile.becquet@gmail.com)
Françoise Pawlotsky (odile.becquet@gmail.com)
Marianne Besnard (odile.becquet@gmail.com)
Micheline Papouin (odile.becquet@gmail.com)
Alexandre Lapillonne (alexandre.lapillonne@nck.aphp.fr)

Version: 2 Date: 6 May 2013

Author's response to reviews:

As requested in your mail received the 4th of May we have added a statement on the Ethics requirements and have modified the contact information.

Sincerely